Author's response to reviews

Title: Laparoscopic transperitoneal lateral adrenalectomy for malignant and potentially malignant adrenal tumors

Authors:

Michal Pedziwiatr (michal.pedziwiatr@uj.edu.pl)
Mateusz Wierdak (wierdakmateusz@poczta.onet.pl)
Maciej Matlok (maciej.matlok@uj.edu.pl)
Magdalena Bialas (mikep@poczta.onet.pl)
Michal Natkaniec (mrrnatk@gmail.com)
Piotr Major (majorpiotr@gmail.com)
Piotr Budzynski (budzynp@wp.pl)
Alicja Hubalewska-Dydejczyk (ahdydej@interia.pl)
Andrzej Budzynski (andrzej.budzynski@uj.edu.pl)

Version: 2
Date: 11 January 2015

Author's response to reviews: see over
Editorial Board,
BMC Surgery

Dear Editors,

We are sending the corrected version of the paper "Laparoscopic transperitoneal lateral adrenalectomy for malignant and potentially malignant adrenal tumors". All changes suggested by the Editor were made.

1. Requesting Ethics Statement - added

2. Conflict of Interest should be changed to Competing Interests - changed

3. Requesting email addresses of all authors in Title Page – added.

4. Please provide Abstract Title – added.

5. Line numbering – added.

Best regards,

Authors.